Author’s response to reviews

Title: Unphosphorylated STAT3 in heterochromatin formation and tumor suppression in lung cancer

Authors:

Pranabananda Dutta (pranabananda@gmail.com)
Lin Zhang (liz004@health.ucsd.edu)
Huijun Zhang (hjzhang021@163.com)
Qin Peng (q3peng@ucsd.edu)
Phillippe R. Montgrain (pmontgrain@ucsd.edu)
Yingxiao Wang (yiw015@eng.ucsd.edu)
Yuanlin Song (song.yuanlin@zs-hospital.sh.cn)
Jinghong Li (jil055@health.ucsd.edu)
Willis Li (wxli@health.ucsd.edu)

Version: 2 Date: 12 Feb 2020

Author’s response to reviews:

1. We added “mice were euthanized by CO2 inhalation per IACUC protocol” to Methods.
2. We added “Mouse xenograft experiments were conducted in accordance with NIH guidelines for the care and use of laboratory animals, and the vertebrate animal protocol has been approved by UCSD Institutional Animal Care and Use Committee (IACUC) (Protocol Number: S13282)” to “Ethics approval and consent to participate”
3. We provided full-length images for Figure 1B in Figure S1.